Author’s response to reviews

Title: Does burst-suppression achieve seizure control in refractory status epilepticus?

Authors:
Kanitpong Phabphal (pkanitpo@medicine.psu.ac.th)
Suparat Chisurajinda (jk.suparat@gmail.com)
Thapanee Somboon (maedajang@hotmail.com)
Kanjana Unwongse (UnnKan@hotmail.com)
Alan Geater (Alan.g@psu.ac.th)

Version: 3 Date: 14 Apr 2018

Author’s response to reviews:

Department of Internal Medicine
Faculty of Medicine
Prince of Songkla University
Hat Yai, Songkhla
THAILAND 90110
Tel: (66) 74-451452
Fax: +66-74-429385
E-mail:
pkanitpo@medicine.psu.ac.th

April 14, 2018

Dear Professor Aidan Neligan

Ref: Reply to Editor’s Comments
Thank you for considering our paper for publication. We have now made a number of changes as the editor has suggested. These are detailed below according to the numbered comments.

1. Consent to Participate:

Please include the name of the IRB that waived the need to obtain consent to participate, or provide a reference to a national legislation that states the need for consent for this type of study is unnecessary.

We have made corrections according to the comment (page 20, line 30-46).

2. Authors' Contributions: This section is still not in line with ICMJE's criteria for authorship as SC, TS and KU have not been accredited with contributing to the drafting or the revising of the manuscript for intellectual content. Please provide more justifications for these authors. If you choose to change your author list you will need to fill out a change in authorship form and send it by email to the Editorial office to be approved by the Editor. The form can be found here:

We have made corrections according to the comment. We have added information in Authors’ contributions (page 20, line 12-15).

3. Figure Legends:

a. Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References.

We have made corrections according to the comment (page 23, line 22-26).

b. Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them.

We have made corrections according to the comment.
Please find enclosed the revised article of the resubmitted manuscript “Does burst-suppression achieve seizure control in refractory status epilepticus?” for your consideration.

Yours sincerely,

Kanitpong Phabphal (correspondence)
Alan Geater
Suparat Chisurajinda
Thapanee Somboon
Kanjana Unwongse